Advanced Software Engineering CSE6340(I) 1

Homeworks #1 & #2

ffl Homework #1: Literature research

. Use syllabus p.4 as your guide . Find 3 periodical/proceeding issues . List related papers for each issue:

– by the three CSE6340 topic areas – total number of papers . Choose your presentation paper . Due: Tuesday, 1/30/96

ffl Homework #2: summary/critique

. Select one paper from homework #1 . Not covered in our compilation . Write a 1-2 page summary/critique . Focus: method, findings, relevance . Due: Tuesday, 2/13/96

Spring, 1997 Prof. Jeff Tian, SMU